Author's response to reviews

Title: Up-Beat UK: A programme of research in to the relationship between coronary heart disease and depression in primary care patients.

Authors:

Andre Tylee (a.tylee@iop.kcl.ac.uk)
Mark Ashworth (markashworth@nhs.net)
Elizabeth Barley (e.barley@iop.kcl.ac.uk)
June Brown (june.brown@kcl.ac.uk)
John Chambers (jboydchambers@aol.com)
Anne Farmer (anne.farmer@kcl.ac.uk)
Zoe Fortune (zoe.1.fortune@kcl.ac.uk)
Mark Haddad (m.haddad@iop.kcl.ac.uk)
Rebecca Lawton (rebecca.lawton@kcl.ac.uk)
Morven Leese (morven.leese@kcl.ac.uk)
Anthony Mann (anthony.mann@kcl.ac.uk)
Anita Mehay (anita.mehay@kcl.ac.uk)
Paul McCrone (paul.mccrone@kcl.ac.uk)
Joanna Murray (Joanna.Murray@kcl.ac.uk)
Carmine Pariante (carmine.pariante@kcl.ac.uk)
Diana Rose (diana.rose@kcl.ac.uk)
Gill Rowlands (rowlang2@lsbu.ac.uk)
Alison Smith (alison.1.smith@kcl.ac.uk)
Paul Walters (p.walters@iop.kcl.ac.uk)

Version: 2 Date: 17 February 2011

Author's response to reviews:

Ethical approval together with REC Ref numbers have been added at appropriate points in the manuscript. All part of the study have received favourable ethical opinions.

I have also uploaded copies of the favourable ethical opinions and a copy of the grant award letter.